# Genetic Diversity of Brazilian *Aedes aegypti*: Patterns following an Eradication Program



# Fernando A. Monteiro<sup>1</sup>, Renata Shama<sup>2</sup>, Ademir J. Martins<sup>3</sup>, Andrea Gloria-Soria<sup>4</sup>, Julia E. Brown<sup>4</sup>, Jeffrey R. Powell<sup>4</sup>\*

1 Laboratorio de Epidemiologia e Sistematica Molecular, Instituto Oswald Cruz—Fiocruz, Rio de Janeiro, Brazil, 2 Laboratório de Biologia Computacional e Sistemas, IOC – Fiocruz, Rio de Janeiro, Brazil, 3 Laboratório de Fisiologia e Controle de Artrópodes Vetores, IOC – Fiocruz, Rio de Janeiro, Brazil, 4 Department of Ecology and Evolutionary Biology, Yale University, New Haven, Connecticut, United States of America

# Abstract

**Background:** Aedes aegypti is the most important vector of dengue fever in Brazil, where severe epidemics have recently taken place. Ae. aegypti in Brazil was the subject of an intense eradication program in the 1940s and 50s to control yellow fever. Brazil was the largest country declared free of this mosquito by the Pan-American Health Organization in 1958. Soon after relaxation of this program, Ae. aegypti reappeared in this country, and by the early 1980s dengue fever had been reported. The aim of this study is to analyze the present-day genetic patterns of Ae. aegypti populations in Brazil.

**Methodology/Principal Findings:** We studied the genetic variation in samples of 11 widely spread populations of *Ae. aegypti* in Brazil based on 12 well-established microsatellite loci. Our principal finding is that present-day Brazilian *Ae. aegypti* populations form two distinct groups, one in the northwest and one in the southeast of the country. These two groups have genetic affinities to northern South American countries and the Caribbean, respectively. This is consistent with what has been reported for other genetic markers such as mitochondrial DNA and allele frequencies at the insecticide resistance gene, *kdr.* 

**Conclusions/Significance:** We conclude that the genetic patterns in present day populations of *Ae. aegypti* in Brazil are more consistent with a complete eradication of the species in the recent past followed by re-colonization, rather than the alternative possibility of expansion from residual pockets of refugia. At least two colonizations are likely to have taken place, one from northern South American countries (e.g., Venezuela) that founded the northwestern group, and one from the Caribbean that founded the southeastern group. The proposed source areas were never declared free of *Ae. aegypti*.

Citation: Monteiro FA, Shama R, Martins AJ, Gloria-Soria A, Brown JE, et al. (2014) Genetic Diversity of Brazilian Aedes aegypti: Patterns following an Eradication Program. PLoS Negl Trop Dis 8(9): e3167. doi:10.1371/journal.pntd.0003167

Editor: Enock Matovu, Makerere University, Uganda

Received April 28, 2014; Accepted August 5, 2014; Published September 18, 2014

**Copyright:** © 2014 Monteiro et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability: The authors confirm that all data underlying the findings are fully available without restriction. All relevant data are within the paper, main text and Supporting Information files.

Funding: This work was supported by the Brazilian Dengue Control Program, Brazilian Ministry of Health and the US NIH grant RO1 Al101112. The funders had no role in the study design, data collection and analysis, decision to publish, or preparation the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

\* Email: jeffrey.powell@yale.edu

# Introduction

Dengue fever is a viral disease transmitted by *Aedes* mosquitoes that occur in tropical and subtropical areas around the world. Due to a widespread distribution, this disease could be more important than malaria in terms of economic impact and morbidity [1-4]. It is estimated that more than two billion people (over 40% of the world's population) are at risk of infection by one or more dengue serotypes [5,6].

Brazil is especially vulnerable to dengue epidemics, with ten times more cases than other Latin American countries during recent outbreaks [6,7]. The main vector of dengue in Brazil is the mosquito *Aedes aegypti*, which is also a vector for yellow fever and chikungunya viruses [8]. *Ae. aegypti* is a particularly adaptable invasive species that has successfully colonized most tropical and subtropical regions of the world. This is due to the vector's highly anthropophilic behavior and ability to lay its desiccation-resistant eggs in man-made water containers, widely available in most developing countries where water distribution and sanitary conditions are rudimentary. Modern transportation and commerce have greatly contributed to the passive geographical spreading of this vector and, consequently, to disease dissemination. Due to the lack of an effective vaccine, currently, dengue control programs rely almost exclusively on vector control efforts [3,9].

Historically, neurotoxic insecticides have been the method of choice to control *Ae. aegypti* populations [10–12]. However, the large-scale unregulated use of insecticides, has exerted intense selective pressures on mosquito populations leading to the development of resistant strains not only in Brazil but also worldwide [13–18]. This undesired outcome increases the need for the creation of new vector control methods. Several emerging technologies are based on various genetic strategies (RIDL, RNAi, HEG, Wolbachia) and are either under development or are

#### **Author Summary**

The mosquito, Aedes aegypti, was historically very important as the major vector of yellow fever, whereas today it is most notorious for being the major transmitter of dengue fever. In the 1940s and 50s, the Pan-American Health Organization organized a campaign to eradicate Ae. aegypti from the New World. They were partly successful, with Brazil being the largest country to be declared free of Ae. aegypti. Within ten years of relaxation of control efforts, Ae. aegypti reappeared in Brazil and today is the vector of the most intense dengue epidemics in the New World. Here, we present population genetic data that are most consistent with the species having truly been eradicated from Brazil rather than simply pushed into small refugia as a consequence of the eradication campaign. The reinfestation most likely resulted from two sources: 1) from northern S. American countries like Venezuela into northwest Brazil and 2) from the Caribbean into the southeast of the country.

already being field-tested [19]. Regardless of the methods employed, knowledge of the genetic variability and population subdivision of mosquito populations is pivotal for the development of rational dengue control programs.

In this context, Brazil is a particularly interesting country regarding dengue epidemiology because it has gone through a well-documented vector eradication program [10,20]. In the first half of the 20<sup>th</sup> century, when dengue was not yet a public health issue, Ae. aegypti populations were widespread and responsible for several yellow fever epidemics, especially in the northeast of Brazil. Motivated by the success achieved by the Anopheles gambiae control program, the Brazilian government launched, in 1947, an initiative to eradicate Ae. aegypti populations based on the use of DDT. In 1958, during the XV Conferencia Sanitária Panamericana in Puerto Rico, Brazil was declared free of Ae. aegypti. The species was again recorded in the late 1970's, probably as a consequence of a reduction in the efficacy of the vector control measures employed [10,20]. The first well-documented outbreak of dengue fever in the country occurred in 1982, in Roraima state, north Brazil [21]. Today, the entire country is endemic for dengue and the last outbreak in 2013 accounted for more than 1.5 million cases (BRAZIL/Health Ministry, 2014). Therefore, dengue fever has become a major public health issue, especially because all four DENV serotypes co-circulate in the country [7].

The first studies to assess the genetic structure of Brazilian Ae. aegypti populations were based upon RAPD markers and revealed high levels of interpopulation genetic differentiation [22,23]. Allozyme-based studies also indicated a high degree of genetic structure and limited gene flow between regions connected by highways and railroads, suggesting that passive mosquito dispersal is not extensive [24,25]. Within cities, such as the densely populated Rio de Janeiro, local genetic differentiation has also been found indicating that this species has extremely limited dispersal capability [26,27].

The analyses of mtDNA sequence data (ND4 and COI) of several Brazilian populations revealed the co-occurrence of two distinct lineages in the country [28,29]. A study of frequencies of the kdr (knock-down resistance) mutations, which confer pyrethroid resistance, found at least three distinct genetic groups in 30 Brazilian populations. [18].

Microsatellites are assumed neutral, highly variable codominant markers commonly used in population genetics. However, they have never been used in a nationwide study of *Ae. aegypti* in Brazil. Here, we present the results of the analysis of 12 microsatellite loci in Brazilian *Ae. aegypti* populations in an effort to better understand the genetic structure of this vector in the country, which may lend insights into the presumed recolonization following eradication events.

#### Methods

#### Sample collection

Ae. aegypti samples were field-collected from 11 sites in Brazil (Table 1). Eggs were collected in multiple ovitraps per locality (to avoid sampling of siblings) and reared to adults for proper taxonomic identification. Samples from generation  $F_0$  up to  $F_2$  were preserved in 70–100% ethanol or dry at  $-80^{\circ}$ C for further analysis. Eight previously studied populations from different countries across South, Central and North America [30] were included in the analyses (Table 1).

#### DNA extraction and amplification

Total genomic DNA was extracted with the DNeasy Kit (Qiagen) following the manufacturer's protocol. Individual genotypes were scored for 12 previously published microsatellite loci [30,31].

#### Data analysis

Microsatellite alleles were scored using Gene Mapper software (Applied Biosystems). The experiments were performed in the Yale Laboratory using the same ABI machine as used by Brown *et al.* [30] and alleles scored in accordance with that publication, so the data presented here are directly comparable to data in Brown *et al.* [30].

To infer the statistical reliability of our markers, each locus was tested for deviations from Hardy-Weinberg expectations on the web version of Genepop v1.2 [32,33]. The same program was used to test all loci pairs for linkage disequilibrium (LD). Markov chain parameters were set at 10,000 dememorizations, 1,000 batches and 10,000 iterations per batch for both HWE and LD. Critical significance levels were corrected for multiple tests using the Bonferroni correction. The probability of null allele occurrence in each locus within each population was calculated using MicroChecker v2.2.3 [34]. When null alleles were found, FreeNA [35] was used to infer the extent of bias imputed by their presence on F<sub>ST</sub> values. Genetic diversity per locus and in each population was estimated by unbiased expected heterozygosity using GenA-LEx v6.5 [36]. The same program was used to compute allele frequencies for all loci across populations and for the Analysis of Molecular Variance (AMOVA). Sample size corrected allelic richness and percentage of private alleles were calculated using HP-Rare v1.0 [37,38]. The software Arlequin v3.5.1.2 [39] was used to compute FST values and their significance between all pairs of populations with 1,000 permutations. Cavalli-Sforza and Edwards distances were computed using the software package Phylip 3.6 [40]. The Cavalli-Sforza distance was chosen since it has been shown to be more robust when null alleles are present [35,41]. Programs of the Phylip package (SEQBOOT, GENE-DIST, NEIGHBOR, CONSENSE) were used to construct a neighbor-joining tree with 1,000 bootstrap replicates. A factorial correspondence analysis (FCA) was performed with the software Genetix v4.0.5 [42] to better analyze the Brazilian samples. Isolation by distance was tested on the IBD web server v3.23 [43] and also through a Mantel test of correlation between geographical (LnKm) and genetic distance matrices ( $F_{ST}/(1-F_{ST})$ ). For both analyses significance was inferred with 1,000 permutations.

The Bayesian approach used in the software STRUCTURE v2.3.2 [44] was used to infer the number of genetic clusters (K) in

Table 1. Population information for the Aedes aegypti samples analyzed.

Population	Region	Gen. in the lab	N*	Year collected	Latitude	Longitude
Aracajú, Sergipe, Brazil	South America	0	24	2010	$-10.909151^{\circ}$	$-37.074454^{\circ}$
Goiania, Goiás, Brazil	South America	0	24	2009	$-16.677715^{\circ}$	$-49.267630^{\circ}$
Maceió, Alagoas, Brazil	South America	1	24	2009	$-9.666252^{\circ}$	$-35.735098^{\circ}$
Mossoró, Rio Grande do Norte, Brazil	South America	1	22	2009	$-5.188036^{\circ}$	$-37.344134^{\circ}$
Pau dos Ferros, Rio Grande do Norte, Brazil	South America	1	16	2009	$-6.102490^{\circ}$	$-38.209222^{\circ}$
Tucuruí, Pará, Brazil	South America	0	19	2010	$-3.769528^{\circ}$	-49.674109°
Marabá, Pará, Brazil	South America	0	48	2011	$-5.369968^{\circ}$	-49.116927°
Natal, Rio Grande do Norte, Brazil	South America	0	47	2010	$-5.794478^{\circ}$	$-35.210955^{\circ}$
São Gonçalo, Rio de Janeiro, Brazil	South America	2	20	2009	$-22.827099^{\circ}$	$-43.054379^{\circ}$
Cachoeiro do Itapemerim, Espirito Santo, Brazil	South America	2	23	2008	$-20.849368^{\circ}$	-41.113221°
Cachoeiro do Itapemerim, Espirito Santo, Brazil	South America	1/2	43	2012/2013	$-20.849368^{\circ}$	-41.113221°
Jacobina, Bahia, Brazil	South America	0	92	2013	$-11.185452^{\circ}$	$-40.536079^{\circ}$
Bolívar, Venezuela	South America	2	48	2004	6.358480°	$-63.580581^{\circ}$
Zulia, Venezuela	South America	2	47	2004	9.967492°	$-72.520483^{\circ}$
Houston, Texas, USA	North America	0	29	2009	29.760196°	$-95.369396^{\circ}$
Coatzacoalcos, Mexico	Central America	0	50	2008	18.149988°	-94.433299°
Pijijiapan, Mexico	Central America	1	47	2008	15.685171°	-93.212254°
Dominica	Caribean	0	48	2009	15.414999°	$-61.370976^{\circ}$
Puerto Rico	Caribean	0	47	2011	18.220833°	$-66.590149^{\circ}$
Miami, Florida, USA	North America	0	47	2010	25.788969°	$-80.226439^{\circ}$

\*Number of individuals collected.

doi:10.1371/journal.pntd.0003167.t001

the whole data set, without prior information of sampling locations. An admixture model was used where alpha was allowed to vary and independent allele frequencies were assumed with lambda set to one. We performed ten independent runs for each value of K (K=1) to the maximum supposed number of populations) with a burn-in phase of 200,000 iterations followed by 600,000 replications. The program Structure Harvester v0.6.93 [45] was used to summarize these results and determine the most likely number of clusters by calculating  $\Delta K$  [46]. Results from STRUCTURE were summarized with the program CLUMPP v1.1.2 [47] and visualized using the program Distruct v1.1 [48]. The program GeneClass2 v2.0 [49] was used for self-assignment tests to infer the degree to which an individual mosquito could be assigned to a specific population. Self-assignment tests were performed with reference populations based on geography and clusters identified by the program STRUCTURE.

#### Results

#### Marker assessment

Although 15 of the 1,244 (1.2%) locus-by-locus tests for LD remained significant after Bonferroni correction, no two loci were consistently correlated across populations. Eleven of the 231 (4.76%)  $F_{IS}$  values deviated significantly from Hardy-Weinberg expectations at the 5% significance level after sequential Bonferroni correction (Table S1). Of the 20 population-specific tests for each marker, zero (AC1, AC2, AC4, CT2, AG5, B2 and B3), one (AG1, AG2 and A1), three (AC5) and five (A9) tests were significant. For A9, all significant tests resulted from an excess of homozygotes, probably due to null alleles as reported in Brown *et al.* [30]. Micro-checker results suggest that locus A9 has a high

probability of having null alleles in 11 populations and AC5 in five. Null allele frequency varied from 0 to 0.32 among populations for the A9 locus and 0 to 0.21 for the AC5 locus (Table S2). Other loci had null allele frequencies predicted as well (in four populations for AG2, three populations for AC1, two populations for B3 and one population for AC4, AG1 and AC2), although none with frequencies >0.14 (Table S2). Null alleles at microsatellite loci are commonly found in insects [50–52] and have been demonstrated to be especially common in species with large population sizes [35], which is likely the case for *Ae. aegypti* populations.

The decrease in diversity caused by null alleles can lead to an overestimation of statistics such as  $F_{\rm ST}$  and identity values [53], especially when there is low gene flow among populations [35,54]. Nevertheless, simulation studies have shown the bias to be small for lower  $F_{\rm ST}$  values and almost none when assignment methods are used [54]. A comparison between FreeNA corrected and non-corrected pairwise  $F_{\rm ST}$  values shows very small deviations in our dataset (Table 2).

#### Levels of genetic variation

Gene frequencies, heterozygosities (H<sub>o</sub> and H<sub>e</sub>), and allelic richness for all loci studied are given in Table S3. All populations have similar diversity measures. AMOVA results show that within population differences account for 83% of the genetic variation found. Private allelic richness was low (N<sub>p</sub><0.08) with only Pau dos Ferros, São Gonçalo, Dominica and Miami with estimates greater than 0.16 (Table S3). Overall F<sub>ST</sub> value (F<sub>ST</sub> = 0.175; 95% confidence interval 0.146–0.204) indicates a moderate level of population differentiation (Table 2). Coatzacoalcos and Houston were the only populations to have higher F<sub>ST</sub> values (ranging from

Table	2. Popt	lation μ	bairwise	F <sub>st</sub> * valu	es for ∕	łe. aegyp	<i>ti</i> populi	ations stu	udied.											
	ARA	0g	МА	MOS	PDF	Ð	MAR	NAT	9G	BOL	zuL	ПОН	COA	ILIA	DOM	PR	MIA	ES08	ES12	JAC
ARA		0.18	0.13	0.03	0.06	0.14	0.12	0.13	0.12	0.20	0.23	0.21	0.34	0.19	0.21	0.16	0.14	0.15	0.15	0.16
99	0.19	,	0.18	0.19	0.13	0.15	0.13	0.13	0.11	0.20	0.19	0.25	0.34	0.22	0.23	0.12	0.13	0.17	0.08	0.17
MA	0.15	0.19	I	0.17	0.10	0.11	0.13	0.09	0.07	0.21	0.24	0.30	0.30	0.22	0.17	0.12	0.16	0.08	0.11	0.10
MOS	0.03	0.20	0.19	ı	0.06	0.14	0.13	0.12	0.12	0.22	0.23	0.23	0.33	0.23	0.21	0.17	0.16	0.16	0.17	0.16
PDF	0.06	0.13	0.11	0.06	T	0.09	60.0	0.06	0.06	0.17	0.14	0.21	0.31	0.16	0.16	0.10	0.08	0.09	0.09	0.09
5	0.15	0.15	0.10	0.15	0.09	ı	0.03	0.12	0.10	0.17	0.13	0.25	0.27	0.22	0.22	0.15	0.10	0.11	0.11	0.15
MAR	0.13	0.13	0.14	0.14	0.10	0.03	ı	0.14	0.10	0.13	0.12	0.21	0.27	0.20	0.20	0.12	0.09	0.12	0.10	0.16
NAT	0.13	0.12	0.09	0.12	0.06	0.11	0.14	ı	0.05	0.19	0.19	0.26	0.31	0.20	0.16	0.11	0.13	0.08	0.07	0.05
SG	0.12	0.11	0.08	0.14	0.06	0.09	0.10	0.06	,	0.13	0.15	0.22	0.31	0.16	0.14	0.06	0.09	0.06	0.04	0.04
BOL	0.20	0.20	0.21	0.22	0.17	0.16	0.13	0.20	0.12	,	0.12	0.19	0.32	0.18	0.20	0.13	0.11	0.19	0.13	0.16
ZUL	0.24	0.19	0.24	0.24	0.15	0.13	0.12	0.19	0.15	0.12	ı	0.24	0.39	0.26	0.23	0.16	0.08	0.21	0.17	0.19
ПОН	0.22	0.26	0.32	0.23	0.22	0.25	0.21	0.26	0.23	0.19	0.24	ı	0.36	0.20	0.21	0.18	0.12	0.33	0.25	0.26
COA	0.35	0.35	0.30	0.33	0.32	0.27	0.27	0.31	0.31	0.32	0.39	0.36	T	0.28	0.29	0.27	0.24	0.31	0.29	0.28
ЫЛ	0.19	0.22	0.22	0.23	0.17	0.21	0.20	0.21	0.16	0.18	0.27	0.21	0.28	,	0.20	0.15	0.13	0.24	0.19	0.18
MOD	0.22	0.24	0.19	0.22	0.18	0.23	0.21	0.17	0.16	0.21	0.24	0.21	0.29	0.21	,	0.12	0.14	0.19	0.17	0.12
PR	0.16	0.12	0.13	0.18	0.11	0.14	0.12	0.12	0.07	0.14	0.16	0.19	0.27	0.16	0.13	ı	0.08	0.14	0.11	0.12
MIA	0.15	0.13	0.16	0.16	0.08	0.09	0.09	0.13	0.10	0.12	0.08	0.12	0.24	0.13	0.15	0.08	,	0.17	0.13	0.14
ES08	0.16	0.17	0.07	0.17	0.09	0.11	0.12	0.09	0.06	0.20	0.21	0.34	0.30	0.24	0.21	0.15	0.17	ı	0.05	0.07
ES12	0.16	0.08	0.11	0.18	0.09	0.10	0.11	0.07	0.04	0.14	0.17	0.26	0.29	0.20	0.18	0.12	0.14	0.05		0.06
JAC	0.16	0.16	0.10	0.16	0.09	0.14	0.16	0.05	0.05	0.17	0.20	0.26	0.29	0.19	0.13	0.13	0.15	0.07	0.06	
*Below Aracajú	diagonal, F – ARA; Goi	i <sub>st</sub> values v ania – GO,	without co Maceió –	rrection, al MA; Mosso	l significar ró – MOS;	ntly differe Pau dos F	nt from zer erros – PDF	o. Above d ; Tucuruí –	iagonal Fre TU; Marabá	eNA correct	ed F <sub>ST</sub> value tal – NAT; S	es. ão Gonçal	o – SG; Bo	livar – BOL;	Zúlia – ZUl	: Houston	- HOU; Coat	tzacoalcos -	. COA; Pijiji	;lll – PlJI;

۲q Dominica – DOM; Puerto Rico – PR, Miami – MIA; Cachoeiro 2008 – ES08; Cachoeiro 2012 – ES12; Jacobina – JAC, doi:10.1371/journal.pntd.0003167.t002

org 4

0.24 to 0.38 in Coatzacoalcos and 0.11 to 0.31 in Houston). Miami and some Brazilian populations had  $F_{ST}$  values lower than 0.10 (Table 2). The genetic distance based NJ tree is reasonably consistent with geographic distances among populations (Figure S1) and was corroborated by Mantel tests of isolation by distance that found significant correlation between the geographical and genetic distance matrices (P<0.001,  $R^2 = 0.53$ ; Figure 1A). When only Brazilian samples were analyzed, weaker isolation by distance was detected by the Mantel tests (P=0.01,  $R^2 = 0.31$ ; Figure 1B).

#### Brazilian population structure and ancestry

A model-based clustering algorithm was used to identify subgroups with distinctive allele frequencies without prior information on population structure. In all analyses, most individuals from the same geographical origin shared similar membership coefficients in inferred clusters. The Evanno et al. [46] method identified K = 2 as the most likely number of clusters, but small peaks on the  $\Delta K$  graph are also apparent at K = 5 and K = 13(Figure S2). The two-cluster analysis groups include all Brazilian populations with Dominica, with the exception of Tucuruí and Marabá (Figure 2). Tucuruí and Marabá are more similar to populations from Venezuela, Mexico, Puerto Rico, and North America. Indeed, the FCA of the Brazilian samples and the NJ tree show that Tucuruí and Marabá (98% bootstrap support; Figures S1 and 3) are very different from all other Brazilian populations. In addition to these two, Mossoró, Aracajú and to some extent Pau dos Ferros also form a slightly differentiated genetic cluster on the FCA analysis (Figure 3).

While the pattern described by the above two genetic clusters is the best supported by the  $\Delta K$  method [46], subtle substructure can be discerned by a more detailed analysis. The five-cluster STRUCTURE plot (Figure 2) shows that most populations have mixed ancestry and only Coatzacoalcos (Mexico) shows a pure genetic composition. In this analysis, the Brazilian samples from Tucuruí and Marabá now group together with Mossoró, Aracajú and, to some extent, Pau dos Ferros, consistent with the FCA analysis that indicates that these last three populations are indeed genetically differentiated as well. The thirteen-cluster plot (Figure 2) further describes the extent of *Ae. aegypti* complex genetic composition in each population. The analysis reflects admixture between groups probably due to recent gene flow among populations, although common ancestry cannot be excluded. The isolation by distance detected among samples also indicates that gene flow occurs between adjacent populations (Figure 1). The thirteen-cluster analysis further separates the Brazilian populations in five distinct clusters (Figure 2), with some mixed ancestry observed, especially in the population proximate to Rio de Janeiro (São Gonçalo), a well-known tourist destination.

Results from GeneClass2 show that when geographical locations were used as the reference populations, 83% of individuals were correctly assigned back to their population of origin. When the number of clusters inferred by STRUCTURE were used, this number increased drastically for K = 2 (94.6%) but not so much for K = 5 (92%) and even less for K = 13 (86.5%), corroborating the higher peak found for K = 2 in the Evanno plot (Figure S2).

Since STRUCTURE seems to identify the higher hierarchy in population differences [46], to better understand the relationships within the two groups identified (Blue and Red in Figure 2, K = 2plot) we performed additional analyses. When the blue group, that encompasses Tucuruí and Marabá with EUA, Mexico, Venezuela and Puerto Rico, is analyzed; the optimal number of clusters determined by the  $\Delta K$  method are K = 2 and K = 8 (Figure S3A and B). At K = 2, the two Mexican populations are differentiated from the rest and display some mixed ancestry with other populations (Figure S4). North America seems to be the most influenced by the Mexican genetic background as was already determined by Brown et al. [30]. Brazilian and Venezuelan populations have less background from Mexico than North America and are, therefore, similar. With K = 8, all populations except the two Brazilian ones seem to be genetically differentiated (Figure S4).

When the red group is analyzed K = 2, K = 3, and K = 5 provide some insights (Figure S5). The two-cluster analysis separates the Brazilian populations from Dominica but a high degree of mixed ancestry can be observed in Jacobina, from the Northeast of Brazil. The three-cluster analysis further differentiates the Brazilian populations showing that Mossoró, Aracajú, Pau dos Ferros and, to some extent, Natal and Maceió group together, although high levels of mixed ancestry can be observed in most



Figure 1. Mantel test of isolation by distance on *Ae. aegypti* populations. Scatter plots of genetic vs. geographical distances (on a logarithmic scale) for pairwise population comparisons. A- All populations (P < 0.001,  $R^2 = 0.53$ ), B- only Brazilian populations (P = 0.01,  $R^2 = 0.31$ ). doi:10.1371/journal.pntd.0003167.g001



**Figure 2. STRUCTURE bar plots for all** *Ae. aegypti* **populations studied.** Individuals are represented by vertical bars along the plot. The height of each color represents the probability of assignment to a specific cluster. The black lines within the plots indicate population limits. Subdivision of all the individuals into K = 2 clusters, subdivision into K = 5 clusters, and into K-13 clusters. Legends below and above apply to all three K plots. doi:10.1371/journal.pntd.0003167.g002

populations (Figure S5). The differentiation of Mossoró, Aracajú and Pau dos Ferros from other Brazilian populations can also be seen on the FCA analysis (Figure 3). The five-cluster analysis further separates Mossoró, Aracajú and Pau dos Ferros in one cluster and shows the geographically close Maceió population to have genetic similarities with Southeastern populations (São Gonçalo and Cachoeiro). Some degree of mixed ancestry can be observed in all populations and this is most apparent in São Gonçalo, Jacobina, Maceió, Pau dos Ferros, Natal, and Cachoeiro (Figure S5). Pau dos Ferros is a small city in the state of Rio Grande do Norte that probably has both the influence of the geographically closer Mossoró and of its state capital, Natal. Interestingly, the two samples from Cachoeiro (2008 and 2012), sampled four years apart, show some degree of differentiation. In a recent study carried out in São Paulo state, Brazil, no differentiation between five sampling years was found [55].



**Figure 3. Factorial Correspondence analysis based on 12 microsatellite loci of** *Ae. aegypti* **populations from Brazil.** Colors correspond to K = 2 cluster analysis displayed on figure 2. doi:10.1371/journal.pntd.0003167.g003



**Figure 4. Collection sites for** *Ae. aegypti* **populations used in this study.** Coloring indicates the two different clusters found on K=2 STRUCTURE analysis (Figure 2). doi:10.1371/journal.pntd.0003167.g004

Despite these subtle genetic patterns, we have strong evidence to conclude that Brazilian populations of *Ae. aegypti* separate into two major genetic groups with distinct affinities to populations outside Brazil as indicated in Figure 4.

## Discussion

Brazil was officially declared free of *Ae. aegypti* in 1958 [20], but reappearance of the species occurred shortly after relaxation of control measures. In its assessment of the efficacy of its eradication program, the Pan American Health Organization (PAHO) admitted that eradication had not been successful in Venezuela, Suriname, Guyana, South USA and a few Caribbean Islands [10]. It is believed that re-colonization of Brazil happened in the 1970's probably from mosquitoes from neighboring countries [7,20].

Our results indicate that two major genetic groups are present in Brazil, one descending from Venezuela and probably other northern American countries and another one from the Caribbean (Figure 4). Bracco et al. [28] using the mitochondrial ND4 gene have also observed two major lineages in Brazil. The first genetic group identified suggests that mosquitoes from Venezuela and possibly the USA have contributed to the northern Brazilian population. Venezuela seems to be an important source of mosquitoes as well as dengue virus serotypes into Brazil [56-58]. Indeed, Silva et al. [59], also using the mitochondrial ND4 gene, have found that populations from the Northern states in Brazil seemed to be similar to those from Venezuela and Peru. In that study, no Caribbean Island was sampled. Venezuelan Ae. aegypti are highly susceptible to DENV2 virus [60] and this could be the reason Lourenço-de-Oliveira et al. [56] have observed that northern Brazilian populations are more susceptible to DENV2 virus than are southern ones. The second genetic group comprises Brazilian southeast and central-west populations and is genetically similar to Dominica in the Caribbean (Figure 2).

Brazil went through a nationwide vector control program based on pyrethroid insecticides from 2001 to 2009. Nevertheless, Linss *et al.* [18] detected three *kdr* genetic groups in the country (North, Northeast and Southeast-Central). Since differential selection pressures acting in the area studied could not account for their findings, the authors argued that the pattern observed could have resulted from genetic differences in the *Ae. aegypti* strains that founded those populations (Linss *et al.* [18]).

In our results, although the most important genetic break occurs between Northern populations and all others (Figures 2 and 4), the FCA also shows that Mossoró, Aracajú and, to some extent, Pau dos Ferros can be differentiated (Figure 3). When a higher cluster number is analyzed on the Bayesian clustering analysis, we see that the same three populations cluster together with the two Northern ones (Figure 2). Other studies of Brazilian Ae. aegypti have identified a genetic break between northern and southern populations [22–24,29,59,61], although the exact location of the break is not always consistent. It is conceivable that the dynamics and mode of inheritance of different genetic markers can account for somewhat different patterns, e.g., cytoplasmic mtDNA versus nuclear genes and neutral genes versus selected alleles such as at insecticide resistance genes (kdr). The isolation by distance found within Brazilian samples suggests some connectivity among populations, so it is not surprising that the two lineages that may have initially re-invaded Brazil are now exchanging genes and perhaps merging.

The origin of these two genetic units seem reasonably clear from our data, although with only a single Caribbean sample (discounting Puerto Rico, considered part of the US) to compare, the origin of the southern lineage is less well established. Bracco et al. [28] suggested that Asia may have been the origin of the southern group, however, they did not sample any Caribbean Islands. Brazil has a long history of international trade within the Americas and Caribbean and only recently has this been shifted to Asian countries. Another indication that indeed Caribbean and not Asian populations might be the source of a Brazilian Ae. aegypti is the fact that Linss et al. [18] have found, in Brazilian populations, the same Caribbean kdr mutation allele, Val1016Ile and not Val1016Gly, that is commonly observed in Asian populations. Furthermore, Brown et al. [62] studying a diverse set on SNPs and nuclear gene sequence data have found that Ae. aegypti probably came from West Africa into the New World, where it dispersed to Asia and Australia. In their study, a Brazilian population from the Southeast (Cachoeiro) is in the same clade as Venezuelan and Caribbean populations, consistent with our findings.

While our data are consistent with the re-colonization hypothesis, we cannot exclude alternatives. The two major genetic groups observed today may have existed prior to 1958; following relaxation of vector control, the expansion from refugia within Brazil could have re-established the pattern present today. However, one expects small refugia to drift to heterogeneous gene frequencies such that subsequent expansion would lead to a mosaic of genetic units not geographically structured. Our data do not support such a scenario. Furthermore, a low genetic diversity would be expected due to a bottleneck period, which was not observed either. Measures of diversity ( $0.39 \le H_o \le 0.67$ ) and allelic richness ( $2.46 \le N_a \le 4.44$ ) are similar in Brazilian samples and other populations from the Americas, even when compared to countries where eradication did not occur (Table S3) [30]. Studies with mitochondrial DNA markers (COI and ND4) have also found high genetic variability in Brazilian samples [28,29]. Thus, while we cannot rule out incomplete eradication, for the reasons stated, recolonization from regions outside Brazil that were never declared free of Ae. aegypti is a simpler explanation consistent with the patterns observed in present day Brazil populations of this vector.

#### **Supporting Information**

**Figure S1** Neighbor-joining tree based on Cavalli-Sforza and Edward's chord distance computed from the allele frequencies of the 12 microsatellite loci analyzed. Unrooted tree. Numbers on branches indicate bootstrap values from 1000 replicates. Colors correspond to K = 2 cluster analysis displayed on figure 2. (TIF)

**Figure S2** Scatter plots of  $\Delta K$  (A) and Log probability of the data (B) for all *Ae. aegypti* populations analyzed.  $\Delta K$  plots are based on the rate of change in the log probability of the data between successive K values.

# (TIF)

**Figure S3** Scatter plots of  $\Delta K$  (A and C) and Log probability of the data (B and D) for the two groups of *Ae. aegypti* populations.  $\Delta K$  plots are based on the rate of change in the log probability of the data between successive K values. A and B – blue group from STRUCTURE analysis on figure 2 (Tucuruí, Marabá, USA, Mexico, Puerto Rico and Venezuela), C and D – red group from STRUCTURE analysis on figure 2 (Dominica and all other Brazilian populations except Tucuruí and Marabá). (TIF)

**Figure S4** STRUCTURE bar plots for *Ae. aegypti* individuals of the blue cluster group from STRUCTURE analysis on Figure 2. Vertical bars along the plot represent individuals. The height of each color represents the probability of assignment to a specific

cluster. The black lines within the plots indicate population limits. Subdivision of the individuals into K = 2 clusters and subdivision into K = 8 clusters. Legends below and above are the same for the two different K plots.

(EPS)

**Figure S5** STRUCTURE bar plots for *Ae. aegypti* individuals of the red cluster group from STRUCTURE analysis on Figure 2. Vertical bars along the plot represent individuals. The height of each color represents the probability of assignment to a specific cluster. The black lines within the plots indicate population limits. Subdivision of the individuals into K = 2 clusters, subdivision into K = 3 clusters and into K-5 clusters. Legends below and above apply to all three K plots.

(EPS)

**Table S1**Aedes aegypti  $F_{IS}$  values by locus.(DOC)

 Table S2
 MicroChecker v2.2.3 null allele frequency.

 (DOC)
 (DOC)

#### References

- Gubler DJ (2002) Epidemic dengue/dengue hemorrhagic fever as a public health, social and economic problem in the 21st century. TRENDS in Microbiology 10: 100–103.
- Gubler DJ (2012) The economic burden of dengue. The American journal of tropical medicine and hygiene 86: 743–744.
- Guzman MG, Halstead SB, Artsob H, Buchy P, Farrar J, et al. (2010) Dengue: a continuing global threat. Nature reviews Microbiology 8: 7–16.
- Bhatt S, Gething PW, Brady OJ, Messina JP, Farlow AW, et al. (2013) The global distribution and burden of dengue. Nature 496: 504–507.
- Halstead SB (2008) Dengue virus-mosquito interactions. Annual review of entomology 53: 273–291.
- WHO (2012) Global Strategy for dengue prevention and control 2010–2012. World Health Organization. Geneva.
- Brathwaite Dick O, San Martin JL, Montoya RH, del Diego J, Zambrano B, et al. (2012) The history of dengue outbreaks in the Americas. The American journal of tropical medicine and hygiene 87: 584–593.
- Consoli R, Lourenço-de-Oliveira R (1994) Principais mosquitos de importância sanitária no Brasil. Rio de Janeiro, Brazil: Ed. Fiocruz. 228 p.
- Maciel-de-Freitas R, Aguiar R, Bruno RV, Guimarães MC, Lourenço-de-Oliveira R, et al. (2012) Why do wee need alternative tools to control mosquitoborne diseases in Latin America? Memórias do Insituto Oswaldo Cruz 107: 828–829.
- Pan American Health Organization (1997) The feasibility of eradicating Aedes agypti in the Americas. Revista Panamericana de Salud Pública 1: 68–72.
- Braga IA, Valle D (2007) Aedes aegypti: inseticidas, mecanismos de ação e resistência. Epidemiol Serv Saúde 16: 277–281.
- 12. Nauen R (2007) Insecticide resistance in disease vectors of public health importance. Pest management science 63:628-633.
- da-Cunha MP, Lima JB, Brogdon WG, Moya GE, Valle D (2005) Monitoring of resistance to the pyrethroid cypermethrin in Brazilian Aedes aegypti (Diptera: Culicidae) populations collected between 2001 and 2003. Memorias do Instituto Oswaldo Cruz 100: 441–444.
- Li X, Schuler MA, Berenbaum MR (2007) Molecular mechanisms of metabolic resistance to synthetic and natural xenobiotics. Annual review of entomology 52: 231–253.
- Montella IR, Martins AJ, Viana-Medeiros PF, Lima JB, Braga IA, et al. (2007) Insecticide resistance mechanisms of Brazilian *Aedes aegypti* populations from 2001 to 2004. The American journal of tropical medicine and hygiene 77: 467– 477.
- Martins AJ, Lins RM, Linss JG, Peixoto AA, Valle D (2009) Voltage-gated sodium channel polymorphism and metabolic resistance in pyrethroid-resistant *Aedes aegypti* from Brazil. The American journal of tropical medicine and hygiene 81: 108–115.
- Bass C, Field LM (2011) Gene amplification and insecticide resistance. Pest management science 67: 886–890.
- Linss JG, Brito LP, Garcia GA, Araki AS, Bruno RV, et al. (2014) Distribution and dissemination of the Val1016Ile and Phe1534Cys Kdr mutations in *Aedes* aegypti Brazilian natural populations. Parasites & Vectors 7: 25.
- McGraw EA, O'Neill SL (2013) Beyond insecticides: new thinking on an ancient problem. Nature reviews Microbiology 11: 181–193.
- Braga IA, Valle D (2007) Aedes aegypti: Histórico de Controle no Brasil. Epidemiol Serv Saúde 16: 113–118.
- Vasconcelos PFC, Rosa APAT, Pinheiro FP, Rodrigues SG, Rosa EST, et al. (1999) Aedes aegypti, Dengue and Re-urbanization of Yellow Fever in Brazil and

**Table S3**Allele frequencies for the 12 microsatellite locianalyzed.

(DOC)

### Acknowledgments

This work is dedicated to the memory of Dr. Alexandre Afranio Peixoto. We thank RM Albano for a critical reading of the manuscript. The authors would also like to thank the Brazilian Dengue Control Program that allowed the utilization of samples collected in the scope of the Brazilian *Aedes aegypti* Insecticide Resistance Monitoring Network (MoReNAa), Brazilian Health Ministry.

#### **Author Contributions**

Conceived and designed the experiments: FAM JRP. Performed the experiments: FAM AGS JEB. Analyzed the data: FAM RS AGS AJM JRP. Contributed reagents/materials/analysis tools: JRP FAM RS AJM. Contributed to the writing of the manuscript: FAM AGS RS AJM JRP.

other South American Countries - Past and Present Situation and Future Perspectives. Dengue Bulletin 23: 1–11.

- Ayres CFJ, Melo-Santos MAV, Solé-Cava AM, Furtado AF (2003) Genetic Differentiation of *Aedes aegypti* (Diptera: Culicidae), the Major Dengue Vector in Brazil. Journal of Medical Entomology 40: 430–435.
- Paduan KS, Araújo-Júnior JP, Ribolla PEM (2006) Genetic variability in geographical populations of *Aedes aegypti* (Diptera, Culicidae) in Brazil elucidated by molecular markers. Genetics and Molecular Biology 29: 391–395.
- 24. Ayres CFJ, Melos-Santos MAV, Prota JRM, Solé-Cava AM, Regis L, et al. (2004) Genetic Structure of Natural Populations of *Aedes aegypti* at the Micro and Macrogeographic Levels in Brazil. Journal of the American Mosquito Control Association 20: 350–356.
- Costa-Ribeiro MCV, Lourenço-de-Oliveira R, Failloux AB (2007) Low Gene Flow of *Aedes aegypti* between Dengue-Endemic and Dengue-Free Areas in Southeastern and Southern Brazil. The American journal of tropical medicine and hygiene 77: 303–309.
- Costa-Ribeiro MCV, Lourenco-de-Oliveira R, Failloux AB (2006) Geographic and temporal genetic patterns of *Aedes aegypti* populations in Rio de Janeiro, Brazil. Tropical medicine & international health : TM & IH 11: 1276–1285.
- Costa-Ribeiro MCV, Lourenço-de-Oliveira R, Failloux AB (2006) Higher genetic variation estimated by microsatellites compared to isoenzyme markers in *Aedes* aegypti from Rio de Janeiro. Memorias do Instituto Oswaldo Cruz 101: 917–921.
- Bracco JE, Capurro ML, Lourenco-de-Oliveira R, Sallum MA (2007) Genetic variability of *Aedes aegypti* in the Americas using a mitochondrial gene: evidence of multiple introductions. Memorias do Instituto Oswaldo Cruz 102: 573–580.
- Scarpassa VM, Cardoza TB, Junior RPC (2008) Population Genetics and Phylogeography of *Aedes aegypti* (Diptera: Culicidae) from Brazil. The American journal of tropical medicine and hygiene 78: 895–903.
- Brown JE, McBride CS, Johnson P, Ritchie S, Paupy C, et al. (2011) Worldwide patterns of genetic differentiation imply multiple 'domestications' of *Aedes aegypti*, a major vector of human diseases. Proceedings Biological sciences/The Royal Society 278: 2446–2454.
- Slotman MA, Kelly NB, Harrington LC, Kitthawee S, Jones JW, et al. (2007) Polymorphic microsatellite markers for studies of *Aedes aegypti* (Diptera: Culicidae), the vector of dengue and yellow fever. Molecular Ecology Notes 7: 168–171.
- Raymond M, Rousset F (1995) GENEPOP (version 1.2): population genetics software for exact tests and ecumenicism. Journal of Heredity 86: 248–249.
- Rousset F (2008) Genepop'007: a complete reimplementation of the Genepop software for Windows and Linux. Molecular ecology resources 8: 103–106.
- Van Oosterhout C, Hutchinson WF, Wills DPM, Shipley P (2004) Microchecker: software for identifying and correcting genotyping errors in microsatellite data. Molecular Ecology Notes 4: 535–538.
- Chapuis MP, Estoup A (2007) Microsatellite null alleles and estimation of population differentiation. Molecular biology and evolution 24: 621–631.
- Peakall R, Smouse PE (2012) GenAlEx 6.5: genetic analysis in Excel. Population genetic software for teaching and research–an update. Bioinformatics 28: 2537– 2539.
- Kalinowski ST (2004) Counting alleles with rarefaction: private alleles and hierarchical sampling designs. Conservation Genetics Resources 5: 539–543.
- Kalinowski ST (2005) Hp-rare 1.0: a computer program for performing rarefaction on measures of allelic richness. Molecular Ecology Notes 5: 187–189.
- Excoffier L, Laval G, Schneider S (2005) Arlequin (version 3.0): An integrated software package for population genetics data analysis. Evolutionary Bioinformatics Online 1: 47–50.

- Felsenstein J (2005) PHYLIP (Phylogeny Inference Package) version 3.6. Department of Genome Sciences, University of Washington, Seattle.
- Chapuis MP, Lecoq M, Michalakis Y, Loiseau A, Sword GA, et al. (2008) Do outbreaks affect genetic population structure? A worldwide survey in *Locusta migratoria*, a pest plagued by microsatellite null alleles. Molecular ecology 17: 3640–3653.
- Belkhir K, Borsa P, Chikhi L, Raufaste N, Bonhomme F (1996–2004) GENETIX 4.05, logiciel sous Windows TM pour la génétique des populations. Laboratoire Génome, Populations, Interactions, CNRS UMR 5000: Université de Montpellier II, Montpellier, France.
- Jensen JL, Bohonak AJ, Kelley ST (2005) Isolation by distance, web service. BMC Genet 6: 13.
- Pritchard JK, Stephens M, Donnelly P (2000) Inference of Population Structure Using Multilocus Genotype Data. Genetics 155: 945–959.
- Earl DA, vonHoldt BM (2012) STRUCTURE HARVESTER: a website and program for visualizing STRUCTURE output and implementing the Evanno method. Conservation Genetics Resources 4: 359–361.
- Evanno G, Regnaut S, Goudet J (2005) Detecting the number of clusters of individuals using the software STRUCTURE: a simulation study. Molecular ecology 14: 2611–2620.
- Jakobsson M, Rosenberg NA (2007) CLUMPP: a cluster matching and permutation program for dealing with label switching and multimodality in analysis of population structure. Bioinformatics 23: 1801–1806.
- Rosenberg NA (2004) Distruct: a program for the graphical display of population structure. Molecular Ecology Notes 4: 137–138.
   Piry S, Alapetite A, Cornuet JM, Paetkau D, Baudouin L, et al. (2004)
- Piry S, Alapetite A, Cornuet JM, Paetkau D, Baudouin L, et al. (2004) GENECLASS2: a software for genetic assignment and first-generation migrant detection. The Journal of heredity 95: 536–539.
- Kobayashi T, Sakurai T, Sakakibara M, Watanabe T (2011) Multiple origins of outbreak populations of a native insect pest in an agro-ecosystem. Bull Entomol Res 101: 313–324.
- Babin R, Fenouillet C, Legavre T, Blondin L, Calatayud C, et al. (2012) Isolation and characterization of twelve polymorphic microsatellite Loci for the cocoa mirid bug *Sahlbergella singularis*. International journal of molecular sciences 13: 4412–4417.
- Ambrose L, Cooper RD, Russell TL, Burkot TR, Lobo NF, et al. (2014) Microsatellite and mitochondrial markers reveal strong gene flow barriers for

Anopheles farauti in the Solomon Archipelago: implications for malaria vector control. International journal for parasitology 44: 225–233.

- Slatkin M (1995) A measure of population subdivision based on microsatellite allele frequencies. Genetics 139: 457–462.
- Carlsson J (2008) Effects of microsatellite null alleles on assignment testing. The Journal of heredity 99: 616–623.
- Campos M, Spenassatto C, Macoris MLG, Paduan KS, Pinto J, et al. (2012) Seasonal population dynamics and the genetic structure of the mosquito vector *Aedes aegypti* in Sao Paulo, Brazil. Ecology and evolution 2: 2794–2802.
- Lourenço-de-Oliveira R, Vazeille M, de Filippis AMB, Failloux AB (2004) Aedes aegypti in Brazil: genetically differentiated populations with high susceptibility to dengue and yellow fever viruses. Transactions of the Royal Society of Tropical Medicine and Hygiene 98: 43–54.
- Figueiredo RMP, Naveca FG, Bastos MS, Melo MN, Viana SS, et al. (2008) Dengue Virus Type 4, Manaus, Brazil. Emerging Infectious Diseases 14: 667– 669.
- Codeço CT, Honório NA, Ríos-Velásquez CM, Santos MCA, Mattos IV, et al. (2009) Seasonal dynamics of *Aedes aegypti* (Diptera: Culicidae) in the northernmost state of Brazil: a likely port-of-entry for dengue virus 4. Memorias do Instituto Oswaldo Cruz 104: 614–620.
- Goncalves da Silva A, Cunha IC, Santos WS, Luz SL, Ribolla PE, et al. (2012) Gene flow networks among American Aedes aegypti populations. Evolutionary applications 5: 664–676.
- Urdaneta-Marquez L, Christopher Bosio FH, Rubio-Palis Y, Salasek M, IV WCB (2008) Genetic Relationships among *Aedes aegypti* Collections in Venezuela as Determined by Mitochondrial DNA Variation and Nuclear Single Nucleotide Polymorphisms. The American journal of tropical medicine and hygiene 78: 479–491.
- Paduan KS, Ribolla PEM (2008) Mitochondrial DNA Polymorphism and Heteroplasmy in Populations of *Aedes aegypti* in Brazil. Journal of Medical Entomology 45: 59–67.
- 62. Brown JE, Evans BR, Zheng W, Obas V, Barrera-Martinez L, et al. (2014) Human impacts have shaped historical and recent evolution in *Aedes aegypti*, the dengue and yellow fever mosquito. Evolution; international journal of organic evolution 68: 514–525.